Author's response to reviews

Title: Spatial distribution of the risk of dengue fever in southeast Brazil, 2006-2007.

Authors:

Ricardo Cordeiro (ricacordeiro@gmail.com)
Maria R Donalisio (rita.donalisio@gmail.com)
Valmir R Andrade (valmir@sucen.sp.gov.br)
Ana CN Mafra (anacarol.nunes@gmail.com)
Luciana B Nucci (lbnucci@terra.com.br)
John C Brown (jcbrown2@ku.edu)
Celso Stephan (celso@stephan.mus.br)

Version: 3 Date: 29 April 2011

Author's response to reviews:

Campinas, April 27, 2011
To BMC Public Health
Editorial Board

Dear Editors,

We have made the finals modification of the manuscript:

• Authors names were referred as asked.
• The Abstract is exactly the same in the manuscript and in the online submission system.
• Equations were adapted to be legible in a 85mm column.
• Reference "24" was included in the text.
• Tables were revised.

I hope that these changes are appropriate. Thanks for your attention. Sincerely,

Maria Rita Donalisio – (Corresponding author)